Author’s response to reviews

Title: Human resource management interventions to improve health workers' performance in low and middle income countries: a realist review

Authors:

Marjolein Dieleman (m.dieleman@kit.nl)
Barend Gerretsen (b.gerretsen@kit.nl)
Gert Jan van der Wilt (G.vanderWilt@ebh.umcn.nl)

Version: 5 Date: 10 April 2009

Author’s response to reviews:

In the title, I have made a typographic change - new title: Human resource management interventions to improve health workers' performance in Low and Middle Income countries: a realist review

On page 15, I have replaced a word:
In the section "Quality improvement" second paragraph: "Interventions included" has been changed to "Study designs included"

On page 22: Authors’ contributions:
I have added "GJvdW advised on the study design"